Author’s response to reviews

Title: Impact of panelists' experience on script concordance test scores of medical students

Authors:

Olivier Peyrony (o.peyrony@hotmail.fr)
Alice Hutin (alice.hutin@aphp.fr)
Jennifer Truchot (jennifer.truchot@aphp.fr)
Raphael Borie (raphael.borie@aphp.fr)
David Calvet (d.calvet@ghu-paris.fr)
Adrien Albaladejo (adrien.albaladejo@gmail.com)
Yousrah Baadj (ybaadj@gmail.com)
Pierre-Emmanuel Cailleaux (pierreemmanuel.cailleaux@aphp.fr)
Martin Flamant (martin.flamant@aphp.fr)
Clémence Martin (clemence.martin@aphp.fr)
Jonathan Messika (jonathan.messika@aphp.fr)
Alexandre Meunier (alexandre.meunier@etu.parisdescartes.fr)
Mariana Mirabel (mariana.mirabel@aphp.fr)
Xavier Treton (xavier.treton@aphp.fr)
Sylvie Chevret (sylvie.chevret@u-paris.fr)
David Lebeaux (david.lebeaux@aphp.fr)
Damien Roux (damien.roux@aphp.fr)

Version: 1 Date: 15 Jun 2020

Author’s response to reviews:

Dear Editor and reviewers,
Thank you very much for your kind comments. We revised the English by a native English speaker (and changed a little bit the title on her advice). We also revised our ethics statement as you suggested.

Please note that there was a mistake in the name of one of the co-authors (Albaladejo instead of Albaladejoa) and that we modified the academic degree of Clémence Martin (PhD).

We hope that this revised manuscript is now more suitable for publication in your journal.

Sincerely